Author’s response to reviews

Title: Adjuncts or adversaries to shared decision-making? Applying the integrative model of behavior to the role and design of decision support interventions in healthcare interactions

Authors:  
Dominick L Frosch (dfrosch@mednet.ucla.edu)  
France Légaré (france.legare@mfa.ulaval.ca)  
Martin Fishbein (mfishbein@asc.upenn.edu)  
Glyn Elwyn (elwyn@cardiff.ac.uk)

Version: 5  Date: 28 September 2009

Author’s response to reviews:

We’ve gone through the copy-edited version and addressed each of the changes requested and also expanded the abstract slightly.

Could you please show us how Table 3 would be added in an appendix, as opposed to included in the paper. We are happy to leave this to your discretion - given that all the materials are online.